# The impact of fiscal decentralization on growth, inflation and inequality in the Americas

Antonio N. Bojanic<sup>1</sup>

### Abstract

This paper analyses the impact of fiscal decentralization on economic growth, inflation and Gini coefficients in 12 countries of the Americas. The findings suggest that the positive impact of this process has been more modest than anticipated, with revenue decentralization having a detrimental effect on economic growth and expenditure decentralization a positive one in developing nations of the Americas. Regarding the impact on income inequality, the results indicate that fiscal decentralization can play an important role in reducing this, particularly on the revenue side, but when decentralization is analysed in developing nations of the Americas only, fiscal decentralization is shown to accentuate rather than mitigate income inequality, which highlights the significant amount of work that is yet to be done before this process delivers on expectations. The findings for the impact of fiscal decentralization on price stability are inconclusive.

#### Keywords

Fiscal policy, tax administration, decentralization in government, economic growth, inflation, income distribution, price stabilization, measurement, North America, Latin America

### JEL classification

E62, H70, O10, O50

#### Author

Antonio N. Bojanic is professor of practice with the Department of Economics at Tulane University. Email: abojanic@tulane.edu.

The author is grateful to James Alm and an anonymous referee for their helpful comments and suggestions. The author alone is responsible for any errors that remain.

## I. Introduction

For most countries in Latin America, the last three decades have been a period of significant devolution of government activities and functions to subnational government structures, particularly at the regional, municipal and local levels. More mature and entrenched democracies with institutions that are increasingly reflecting the demands of specific constituencies, a desire for greater representation of different subgroups and regions within societies, and the realization that there may be efficiency gains from transferring responsibilities to the areas and communities most directly affected by government interventions are among the reasons for this trend towards decentralization of government activities.

The main subject of this study is fiscal decentralization, which occurs when central governments transfer certain revenue and expenditure responsibilities to subnational levels of government. Economic research on this topic has for the most part concentrated on how fiscal decentralization impacts governance and economic growth, but more recent efforts have begun analysing the ways in which it affects poverty, income distribution and fundamental rights, including civil and political rights as well as economic freedom.

Although current research has extensively studied the impact of decentralization on growth, no clear-cut conclusions have emerged, particularly when the process of decentralization is analysed from the perspective of developing nations. The lack of conclusive evidence is even more evident when the impact of decentralization on other variables such as income distribution and economic stability is examined. This paper aims to help bridge some of the gaps in current research by analysing how fiscal decentralization has impacted economic growth, inflation and income inequality in a region of the world where the issue of decentralization has been at the forefront of structural reforms for several decades.

The rest of the paper is organized as follows. Section II presents a brief review of the literature on the relationship between fiscal decentralization, growth, economic stability and inequality. Section III introduces the theoretical arguments for the expected impact of decentralization on the principal variables of interest. Section IV presents the data and methodology and section V the analysis of the empirical results. Section VI summarizes the principal findings and analyses their policy implications.

### II. Literature review

One of the main areas of economic research into fiscal decentralization deals with its effects on economic growth. Several studies have carried out cross-country and country-specific analyses of the issue, mostly but not exclusively in developed nations. A related theme is the way decentralization has affected economic stability, and there has also been significant research on this topic. The most recent areas of interest are the impacts on poverty, income inequality and fundamental human liberties. A brief review of some of the most important works on this array of topics is presented below.

Cross-country studies of the way fiscal decentralization impacts economic growth are many. Important early contributions include Davoodi and Zou (1998), who worked with data for 46 countries and found a negative correlation between fiscal decentralization and growth in developing nations, but none in developed economies. Martínez-Vázquez and McNab (2003) concluded that while fiscal decentralization might in fact have an impact on growth, the theoretical underpinnings for this relationship remained underdeveloped and hence no definite answer could be provided. Martínez-Vázquez and McNab (2006) found that when a negative correlation between decentralization and growth was established for developed countries, it could be offset by the positive impact of decentralization on macroeconomic stability. Thornton (2007) worked with data for 19 countries of the Organization for Economic Cooperation and Development (OECD) and found that when fiscal decentralization was measured only by the revenues over which subnational governments had full autonomy, its impact on economic growth was not statistically significant.

More recent works include Rodríguez-Pose and Ezcurra (2011), reviewing a set of 21 OECD countries and finding a significant negative association between fiscal decentralization and economic growth, despite the inclusion of several control variables and adjustments to account for differences in expenditure preferences by subnational governments. Amagoh and Amin (2012) concluded that while there might be benefits from fiscal decentralization, its impact on growth was constrained by a number of factors that depended on the contexts of the societies involved. Baskaran and Feld (2013) found, also for a set of OECD countries, that fiscal decentralization appeared to have a statistically insignificant negative effect on growth when proxied by standard indicators of the Government Finance Statistics (GFS) type, but a statistically significant negative impact when new indicators reflecting the degree of subnational tax autonomy were used. Gemmell, Kneller and Sanz (2013) found for a set of OECD countries that spending decentralization had tended to be associated with lower economic growth and revenue decentralization with higher growth. Blöchliger (2013) found a positive association between fiscal decentralization and GDP per capita in OECD countries, with revenue decentralization having a greater impact than spending decentralization. Representative works for individual countries include Xie, Zou and Davoodi (1999) for the United States, Yifu Lin and Liu (2000) for China and Rao (2000) for India.

On the issue of economic stability and the way fiscal decentralization affects it, representative works include Neyapti (2004), which takes a set of countries with varying levels of inflation and finds that revenue decentralization has a negative impact on inflation in higher-inflation countries if accompanied by both central bank independence and local accountability, while in lower-inflation countries the negative impact on inflation affectors; Neyapti (2010), which analyses the topic of fiscal discipline and concludes that for a set of 16 countries expenditure and revenue decentralization reduces budget deficits; Rodden, Eskeland and Litvack (2003), which analyses the issue of how fiscal discipline is maintained when lower levels of government take responsibility from national authorities and examines how "hard" and "soft" budget constraints impact economic stability in countries with varying degrees of political and institutional development; and Jalil, Harun and Che Mat (2012), which focuses on price stability for 62 countries and finds that decentralization appears to lower the inflation rate to an extent that depends on the level of corruption in political institutions. Country-specific studies include Bodman and others (2009) for Australia, Iqbal and Nawaz (2010) for Pakistan and Okonkwo and Godslove (2015) for Nigeria.

Concerning fiscal decentralization, poverty and income distribution, significant contributions include Boex and others (2006), which in addition to providing a comprehensive survey of the literature on the topic offers a set of qualitative suggestions for conducting decentralization reforms from a propoor perspective; Sepúlveda and Martínez-Vázquez (2011), which takes a large dataset of countries and finds that fiscal decentralization appears to reduce poverty as long as the share of subnational expenditures is no greater than one third of total government expenditures, and also to reduce income inequality, but only if general government represents a significant share of the economy; Goerl and Seiferling (2014), which takes a large dataset of countries and finds that the decentralization of government expenditures can help achieve a more equal distribution of income if a number of conditions are met; and Sacchi and Salotti (2014), which looks at a set of OECD countries and finds that a higher degree of tax decentralization is associated with higher household income inequality. At the individual country level, a sample of works includes Moon (2003) for South Korea, Song (2013) for China and Cavusoglu and Dincer (2015) for the United States.

A recent area of research involves analysis of the impact of fiscal decentralization on what can be described as fundamental human rights, a term that encompasses both civil and political rights and economic freedom. Although this line of research is not new in other areas of the social sciences, as can be seen in the early contribution by Kaufman (1969) and a great many subsequent articles, such as Michels (2011) and Islam (2015), it has been little explored in economics. Notable exceptions include Weingast (2009) and Bojanic (2016).

# III. The theoretical foundations for the relationship between fiscal decentralization, growth, inflation and income inequality

Analysis of current research on fiscal decentralization and its impact on a range of indicators leaves an impression of uncertainty about how decentralization will affect variables such as growth, economic stability and income inequality. However, the fact that empirical work has not provided a clear picture in its current state has not prevented economists from hypothesizing about the ways in which decentralization is expected to affect these very variables. The most recent theoretical work will be briefly summarized here and an attempt made to highlight the issues that are likely to play an important role in our understanding of how fiscal decentralization affects growth, price stability and income distribution in the Americas.

Concerning the potential growth impact of fiscal decentralization, there is already a significant body of theoretical work on the subject (see, for instance, Oates, 1993; Brueckner, 2005; Martínez-Vázquez and McNab, 2006), and the answer seems to hinge on whether a central authority is best able to utilize fiscal policy to attain long-term growth, or whether a decentralized structure for administering public funds is more capable of delivering outcomes that will translate into growth. While most researchers seem to agree that a positive correlation between decentralization and growth should be expected, owing to better targeting of growth-enhancing infrastructure and greater incentives to save in decentralization is taking place matters. The inference, then, is that although a positive correlation between fiscal decentralization and economic growth is expected, the state of development of an economy will determine whether the decentralization process is able to resolve into policies that generate growth over time.

As regards the impact of fiscal decentralization on inflation, Martínez-Vázquez and McNab (2006) and Treisman (2000), among others, have developed a theoretical framework for the ways in which decentralization is likely to affect price stability. Without hypothesizing about the specific direction in which price stability is likely to be affected by decentralization, their empirical work tends to show that in (mostly) developed economies there is an inverse correlation between inflation and fiscal decentralization, implying that lower inflation levels are more likely in those nations with more decentralization may actually generate higher inflation. From the perspective of countries in the Americas, an important consideration is that one of the principal reasons for the very high inflation rates they have experienced over time is unrestrained government expenditure, with a significant percentage of this occurring at lower levels of government. This draws attention to the very real concern that devolving this specific function to subnational levels of government may once again foster inflationary pressures.

Theoretical work on the way income distribution is affected by fiscal decentralization is not as developed as that for growth and economic stability. An important exception is Beramendi (2003), which offers a theoretical model for analysing how decentralization interacts with the politics of redistribution and inequality and argues that decentralization in itself does not necessarily lead to higher (or lower) levels of income inequality, but rather inequality is to a large extent a function of regions' internal social and political structures. Empirical studies are more numerous and include, among others, Durham (1999), Sepúlveda and Martínez-Vázquez (2011) and, more recently, Goerl and Seiferling (2014). From the perspective of this article, if growth is assumed to be a necessary but not sufficient condition for any increase in income equality, as argued by Kuznets (1995), then fiscal decentralization, to the extent that it is expected to have a positive impact on growth, should also eventually bring greater income equality.

Coupled with the theoretical and empirical findings just described, two important additional factors to consider regarding decentralization in developing nations of the Americas are the limited institutional

ability of subnational levels of government to collect their own revenues and the very real economic, political and cultural disparities that exist within and between countries.

An inability to collect their own revenues makes subnational governments dependent on central government transfers. This situation creates inefficiencies, as these transfers may not be automatic and may be tied to political calculations. The regional disparities within and between countries mean that decentralization in the developing part of the American hemisphere has taken place in very heterogeneous settings, and hence it should not be surprising that the degree of decentralization varies not only between countries but also between regions within each country. Issues of income inequality, the degree of urbanization, territorial imbalances and literacy rates are but a few of the factors that may affect how the process of decentralization is able to take hold in a particular setting. The point is that in the context of decentralization in this part of the world, regional disparities and the dependence of subnational governments on central government for tax collection are likely to play a significant role in the effectiveness and usefulness of fiscal decentralization.

# IV. Data and methodology

One of the most significant challenges for a cross-country study of fiscal decentralization is properly measuring the extent of decentralization in several layers of government. A related issue when the study focuses on (mostly) developing nations is the difficulty of finding reliable and credible data. An optimal scenario would be one in which the dataset constructed was fully comparable across countries and truly reflected the autonomous decisions of subnational governments. As might be expected, constructing such a dataset is a formidable undertaking, not least because it requires knowledge of the degree of autonomy of subnational governments over revenue collection and expenditure decisions. It also calls for a thorough understanding of each nation's tax system, and particularly the structure of revenue-sharing between regions, the nature of grants and transfers between the central government and subnational levels of government, and the overall level of regional political autonomy. Given the difficulty of finding decentralization indicators that successfully identify all these, the standard practice in the economics literature has been to utilize data collected by the International Monetary Fund (IMF) and reported in its Government Finance Statistics Yearbook (GFSY)<sup>2</sup> as the primary source for revenue and expenditure data at national (general) and subnational levels of government. Although GFSY does not report the nature of government transfers or identify whether transfers and grants are under the control of the national or subnational levels, and indeed does not currently have disaggregated data for many developing nations, it is also the primary data source for the present study, albeit not the only one, since revenue decentralization data from OECD,<sup>3</sup> the Economic Commission for Latin America and the Caribbean (ECLAC)<sup>4</sup> and, when possible, the national institutes of statistics or comparable government institutions of each country have been used in addition to those of the GFSY.<sup>5</sup>

The standard measures of fiscal decentralization utilized in most decentralization studies are the ratio of total subnational government revenues to general government revenues and the ratio of total subnational government expenditures to general government expenditures. These two fiscal decentralization indicators are also used here. The GFSY, OECD, ECLAC and national data provide

<sup>&</sup>lt;sup>2</sup> See [online] http://www.imf.org/external/pubs/ft/gfs/manual/comp.htm.

<sup>&</sup>lt;sup>3</sup> See [online] www.oecd.org/ctp/federalism/fiscal-decentralisation-database.htm.

<sup>&</sup>lt;sup>4</sup> See [online] http://estadisticas.cepal.org/cepalstat/portada.html?idioma=english.

<sup>&</sup>lt;sup>5</sup> Decentralization data for the Plurinational State of Bolivia are easily obtainable from the country's National Institute of Statistics (www.ine.gob.bo). Likewise, data on Argentina can be obtained from the Federal Tax Commission of Argentina (www.cfi.gov.ar) and from the Ministry of Treasury and Public Finance of Argentina (www.economia.gob.ar). For the rest of the developing countries of the Americas in this study, obtaining decentralization data from a national government entity was more challenging, so use was made of the data reported by one or more of GFSY, OECD and ECLAC.

information at the consolidated general government level and, for some countries, at the regional, state and local government levels. Revenues (expenditures) at subnational levels of government (regional, state and local) were added together to come up with a single figure for subnational government revenues (expenditures). Of the 23 countries in the Americas (excluding the Caribbean), data disaggregated between the general and subnational levels of government are available for 12 nations, and this study accordingly focuses on this subsample for which data are available.<sup>6</sup> Yearly observations run from 1972 to 2015, although the dates of the data available for the 12 countries do not necessarily coincide. Depending on (i) whether the revenue or expenditure decentralization indicators are used as regressors, (ii) the specific methodology utilized in estimating a regression and (iii) the dependent variable of the model (per capita GDP growth, the inflation rate or the Gini coefficient), the number of observations ranges from a low of 91 to a high of 208. The end result is an unbalanced panel dataset with a maximum of 208 observations for 12 countries of the Americas running from 1972 to 2015. Although there are significant gaps in the dataset, it was decided that no averages or linear approximations should be used to fill in the gaps, with the actual dataset instead being allowed to speak for itself.

The three dependent variables utilized here are per capita GDP growth,<sup>7</sup> the inflation rate and the Gini coefficient. The control variables for the set of regressions relating to fiscal decentralization and economic growth are the inflation rate (expressed as a percentage); gross domestic savings (percentage of GDP) as a proxy for capital formation; openness to international trade ((exports + imports)/GDP, expressed as a percentage); remittances (percentage of GDP); foreign direct investment (FDI) (percentage of GDP); unemployment rate (percentage); general government final consumption expenditures (percentage of GDP) as a proxy for the size of government; urban population (percentage); and a political and civil liberties ratio as a measure of political stability and basic rights.<sup>8</sup> With the inflation rate as the dependent variable, control variables include GDP per capita in levels (at purchasing power parity, in logs), openness to international trade, general government final consumption expenditures, military expenditures (as a percentage of GDP), FDI and remittances. Finally, when the Gini coefficient is the dependent variable, the control variables are GDP per capita in levels and GDP per capita squared, to take account of what Kuznets (1955) hypothesizes about per capita income growth initially increasing inequality but eventually reducing it; urban population; openness to international trade; a political and civil liberties ratio to attempt to capture the extent to which basic human rights affect inequality; remittances; gross domestic savings; general government final consumption expenditures; the inflation rate; Internet users (per 100 people); and the unemployment rate.<sup>9</sup>

The particular specifications for each case conform to previous research on similar topics, but additional control variables have been included where deemed pertinent, namely remittances and a political and civil liberties ratio when economic growth is the dependent variable, remittances when it is the inflation rate, and a political and civil liberties ratio, Internet users and the unemployment rate when it is the Gini coefficient.

Where the model specification is concerned, different regression methodologies were used to deal with information limitations and gaps within an unbalanced panel dataset, the likely correlation of observations within and across sections and the very wide variability of data for the countries in the sample. Specifically, generalized least squares (GLS) and instrumental variables regressions were estimated to

<sup>&</sup>lt;sup>6</sup> The 12 countries included in this study are Argentina, Brazil, Canada, Chile, Colombia, Costa Rica, El Salvador, Mexico, Paraguay, Peru, the Plurinational State of Bolivia and the United States.

<sup>&</sup>lt;sup>7</sup> Figures for per capita GDP growth are calculated in purchasing power parity.

<sup>&</sup>lt;sup>8</sup> The source for this combined ratio is Freedom House, which compiles separate indices of political rights and civil liberties and prepares qualitative assessments of the degree of liberty in each country. Here, the two indices and the qualitative assessments have been combined into a single index of political and civil liberties ranging from 0.18 (most free) to 1.00 (least free). See [online] http://freedomhouse.org/report-types/freedom-world#.VY\_fWI1RHcw.

<sup>&</sup>lt;sup>9</sup> Excepting political and civil liberties, the source for all variables is the World Bank. See [online] http://data.worldbank.org/datacatalog/world-development-indicators.

allow for cross-sectional and intrasectional heteroskedasticity and autocorrelation,<sup>10</sup> as were GLS with fixed and random cross-sectional effects to allow, respectively, for omitted variable bias and for the impacts of time-invariant variables. Additionally, generalized method of moments (GMM) regressions were estimated to reflect the dynamic nature of the relationship between fiscal decentralization, growth, inflation and income inequality.<sup>11</sup>

# V. The results of the empirical analysis

As an introduction to the empirical analysis, table 1 reports summary statistics for the 12 countries of the Americas analysed in this study, including the time period covered by each decentralization indicator.

	(		/					
Country	Subnation of ge	al government eneral governn	revenues as pro nent revenues <i>(</i> %	oportion %)	Subnatior of g	nal governmen eneral governr	t expenditures as ment expenditure	s proportion es <i>(%)</i>
Country	Period	Average whole period	Highest (year)	Lowest (year)	Period	Average whole period	Highest (year)	Lowest (year)
Argentina	1990, 2000, 2005, 2007-2015	37.35	45.59 (2015)	21.17 (2005)	1980-2013	42.52	51.51 (2006)	24.05 (1982)
Bolivia (Plurinational State of)	1985-2014	31.05	46.17 (2003)	15.14 (1985)	1986-2014	27.57	36.98 (1997)	14.95 (1986)
Brazil	2000-2014	54.01	62.47 (2014)	49.19 (2005)	2006-2012	52.34	54.73 (2008)	50.76 (2009)
Canada	1979-2014	72.59	78.10 (2010)	69.35 (1989)	2000-2014	73.30	77.59 (2014)	68.19 (2000)
Chile	1974-2014	8.56	14.06 (2002)	2.75 (1974)	2000-2001	13.35	13.50 (2000)	13.20 (2001)
Colombia	1990, 1998-2003, 2005, 2010, 2014	33.94	46.39 (1999)	20.56 (2005)	1998-2000	39.33	40.16 (2000)	38.76 (1998)
Costa Rica	2000-2014	5.01	6.95 (2007)	3.19 (2000)	2002-2007	3.98	6.35 (2005)	3.08 (2002)
El Salvador	2002-2010	7.61	10.11 (2009)	4.54 (2002)	-	-	-	-
Mexico	1972-2013	29.63	44.76 (2009)	18.96 (1989)	1990-2013	44.58	59.66 (1997)	27.60 (1990)
Paraguay	2005-2012	8.92	10.86 (2012)	7.73 (2010)	2005-2012	9.34	11.17 (2012)	8.02 (2010)
Peru	1995-2012	30.42	39.19 (2009)	21.50 (1998)	1995-2012	30.00	40.88 (2012)	21.08 (1998)
United States	1980-2014	56.13	60.77 (1992)	51.48 (2013)	1990-2014	51.50	59.85 (2000)	46.27 (2011)

 Table 1

 The Americas (12 countries):<sup>a</sup> decentralization indicators and summary statistics

Source: Prepared by the author on the basis of data from International Monetary Fund (IMF) [online] http://data.imf. org/?sk=E86E9088-3830-4CA3-B240-1B0EC5E15221; Organization for Economic Cooperation and Development (OECD) [online] www.oecd.org/ctp/federalism/fiscal-decentralisation-database.htm; Economic Commission for Latin America and the Caribbean (ECLAC) [online] http://estadisticas.cepal.org/cepalstat/portada.html?idioma=english; Federal Tax Commission of Argentina [online] www.cfi.gov.ar; Ministry of Treasury and Public Finance of Argentina [online] www.economia.gob.ar; and National Institute of Statistics of the Plurinational State of Bolivia [online] www.ine.gob.bo.

<sup>a</sup> Argentina, Brazil, Canada, Chile, Colombia, Costa Rica, El Salvador, Mexico, Paraguay, Peru, the Plurinational State of Bolivia and the United States.

The summary statistics demonstrate that there is a great deal of variation between the countries of the Americas in their degree of fiscal decentralization. Of all the countries included in the analysis, Canada is the most decentralized, with both the revenue and expenditure indicators averaging well over 70%.<sup>12</sup> Costa Rica, El Salvador and Paraguay are at the other end of the spectrum, with average

<sup>&</sup>lt;sup>10</sup> With both the GLS and instrumental variables regressions, the Prais-Winsten panel-corrected standard error (PCSE) within estimator was estimated to correct for serial correlation (cross-sectional and between-period correlation).

<sup>&</sup>lt;sup>11</sup> The GMM regressions were estimated using the PCSE method to take account of cross-sectional and between-period correlation.

<sup>&</sup>lt;sup>12</sup> As defined in section IV, the revenue (expenditure) decentralization indicator is the ratio of subnational government revenues (expenditures) to general government revenues (expenditures).

decentralized revenues and expenditure alike in the single digits.<sup>13</sup> Between these two extremes, the degree of fiscal decentralization in the rest of the countries is not homogeneous, with nations like Brazil and the United States showing significant degrees of decentralization (revenue and expenditure decentralization indicators at around the 50% level) that do not however approach that in Canada. In the remaining group of countries, Argentina and Colombia seem to tilt towards greater decentralization, with average revenue and expenditure percentile indicators in the high thirties or low forties, while Mexico, Peru and the Plurinational State of Bolivia fall somewhere in between, with both indicators averaging around 30%.<sup>14</sup> Chile leans to a lower level of decentralization, with revenue decentralization in the high single digits and expenditure decentralization in the low teens. It is also noteworthy that while the raw data seem to suggest an increasing degree of decentralization for most countries as time passes, this is not true for all, as demonstrated by the United States, where both revenue and expenditure decentralization indicators seems to show a downward trend over time.

Regression results showing the impact of fiscal decentralization on GDP per capita, inflation and the Gini coefficient are reported in two sets, the first including all 12 countries and the second excluding Canada and the United States, the two nations with arguably the most developed and stable decentralization regimes in the hemisphere. Excluding these two nations has the benefit of showing the fiscal decentralization situation from the perspective of developing countries of the Americas only.

The impact of fiscal decentralization on growth is analysed first. Table 2 presents regression results when all 12 countries are included and the dependent variable is per capita GDP growth.

The five columns in table 2 report estimates for the five methodologies described in section IV. In the first, a GLS model reflecting cross-sectional and intrasectional heteroskedasticity and autocorrelation is presented. The second and third columns report GLS models with fixed and random cross-sectional effects. The fourth and fifth columns report estimates for when the variables are instrumented and when GMM is used. The instrumental variables and GMM specifications were estimated in consideration of both cross-sectional and intrasectional heteroskedasticity and autocorrelation. Additionally, each column contains two regression results: the first shows estimates for when the fiscal decentralization indicator is based on revenue and the second for when it is based on expenditure.

As is evident from the results reported, the revenue-based fiscal decentralization indicator consistently shows a negative impact on economic growth. The coefficients for this variable in all specifications are negative, of approximately equal size, and statistically significant in all cases, excepting the GLS specification with fixed effects. On the expenditure side, the fiscal decentralization variable is mostly positive but only statistically significant when estimated with GMM, which is consistent with the assumptions made in section III, particularly the supposition that decentralized expenditures were more likely to be targeted at growth-enhancing investment projects.<sup>15</sup> Taken as a whole, however, the empirical evidence does not conclusively support the hypothesis that decentralization is conducive to growth, particularly where revenue decentralization is concerned.

<sup>&</sup>lt;sup>13</sup> In El Salvador, the level of decentralization is only this low on the revenue side.

<sup>&</sup>lt;sup>14</sup> In Mexico, the degree of decentralization on the expenditure side is quite high, averaging around 45% for the 1990-2013 period. On the revenue side, the indicator averages only around 30%, but this may reflect the longer time period (1972-2013).

<sup>&</sup>lt;sup>15</sup> When the specification is instrumented, the expenditure decentralization variable is found to be negative and statistically significant at the 10% level, drawing attention to the weakness of the statistical evidence found in this study to support the expected positive impact of fiscal decentralization on economic growth.

	The	e Americas (1	2 countries	):ª fiscal dec	entralizatio	n and GDP p	er capita			
	Generalized leand of the cross allow for cross intrasectional he and autocor effects spe	ast squares to sectional and teroskedasticity relation (no cification)	Generalized Ir with cross-se effects and	east squares cctional fixed no weights	Generalized I with cross-sec effects and	east squares tional random no weights	Instrumental allow for cross- intrasectional het and autocorr effects spe	variables to sectional and teroskedasticity elation (no cification)	Generalized meth	od of moments
Decomposition (0/)	-0.057**		-0.066		-0.056**		-0.067***		-0.073**	
Kevenue aecentralization (%)	(2.002)		(0.057)		(0.024)		(0.017)		(0.037)	
Econditure decentralization 707		0.008		0.120		0.001		-0.045*		0.160***
באטפוומונעו פ מפכפוונו מווצמנוטוו (%)		(0.019)		(0.084)		(0.031)		(0.024)		(0.049)
	0.001	-0.005***	0.001	-0.005***	0.001	-0.004***	-0.016	-0.001	-0.004	-0.006***
IIIIIauuii Tale ( <i>%)</i>	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.012)	(0.003)	(0.006)	(0.001)
Construction consistent (10) of COD	0.295***	0.208***	0.304***	0.148	0.247***	0.148	0.257***	0.242***	0.526***	0.367***
aioss autitestic savirigs (% ur aur)	(0.043)	(0.058)	(0.092)	(0.169)	(0.069)	(0.106)	(0.052)	(0.077)	(0.064)	(0.119)
Openness to international	-0.061**	-0.010	-0.022	-0.005	-0.051 **	-0.001	-0.048***	-0.005	-0.226***	-0.111*
trade ((X + M)/GDP) (%)	(0.013)	(0.014)	(0.049)	(0.093)	(0.020)	(0.025)	(0.015)	(0.017)	(0.025)	(0.066)
	0.306***	-0.204	0.044	-0.494	0.285**	-0.289	0.228***	-0.062	0.433*	-0.777***
Neilliudices (% or anr)	(0.067)	(0.190)	(0.359)	(0.487)	(0.118)	(0.336)	(0.084)	(0.262)	(0.258)	(0.249)
Earnian direat incommut 10/ of 000	-0.055	-0.016	0.121	-0.325	-0.080	-0.088	-0.345***	-0.080	0.133	-0.573***
	(0.070)	(0.077)	(0.143)	(0.203)	(0.110)	(0.147)	(0.117)	(0.116)	(0.088)	(0.079)
nomnlaumont_rato_/0/_of_tatol_labour_forcol	-0.203**	-0.220***	-0.232**	-0.443***	-0.274***	-0.327***	-0.143*	-0.150*	-0.372***	-0.476***
	(0.057)	(0.054)	(0.120)	(0.138)	(0.089)	(0.091)	(0.085)	(0.079)	(0.071)	(0.072)
General government final consumption	0.045	-0.266***	-0.003	-0.574***	0.073	-0.293**	0.025	-0.095	-0.858***	-0.726***
expenditure (% of GDP)	(0.062)	(0.073)	(0.210)	(0.197)	(0.114)	(0.114)	(0.075)	(0.116)	(0.175)	(0.072)
Ithor controlition (0/ of total)	-0.067**	-0.019	-0.155	-0.051	-0.042	0.004	-0.055*	-0.011	-0.043	0.181
Ulball population ( <i>% of total</i> )	(0.029)	(0:030)	(0.129)	(0.317)	(0:039)	(0:056)	(0.032)	(0.044)	(0.058)	(0.178)
Dolition and airil libortion ratio	-4.348**	-2.979*	4.911	-1.406	-2.351	-3.449	-5.218***	-7.611***	7.834***	-3.626
רטווונכמו מווט כואון ווטפו נופא ומנוס	(1.300)	(1.840)	(3.165)	(4.604)	(1.839)	(2.950)	(1.653)	(2.484)	(1.126)	(2.935)
Observations	208	158	208	158	208	158	187	147	175	136
Adjusted R <sup>2</sup>	0.31	0.37	0.14	0.21	0.11	0.18	0.27	0.30		

01:000 relization and GDP ner יייייי איייייי Table 2 01/000000

Source: Prepared by the author on the basis of data from International Monetary Fund (IMF) [online] http://data.imf.org/?sk=E86E9088-3830-4CA3-B240-1B0EC5E15221; Organization for Economic Cooperation and Development (OECD) [online] www.oecd.org/ctp/federalism/fiscal-decentralisation-database.htm; Economic Commission for Latin America and the Caribbean (ECLAC) online] http://estadisticas.cepal.org/cepalstat/portada.html?idioma=english; Federal Tax Commission of Argentina [online] www.cfi.gov.ar; Ministry of Treasury and Public Finance of Argentina online] www.economia.gob.ar; National Institute of Statistics of the Plurinational State of Bolivia [online] www.ine.gob.bo; Freedom House [online] http://freedomhouse.org/report-types/ (\*) = significant at 10%, (\*\*) = significant at 5% and (\*\*\*) = significant at 1%. The dependent variable is per capita GDP growth. All regressions include an intercept term (not shown in table). freedom-world#.VY\_M11RHcw; and World Bank [online] http://data.worldbank.org/data-catalog/world-development-indicators.

The behaviour of the control variables is also noteworthy. With most, the results conform to expectations about their likely impact on growth. For instance, savings have the expected positive impact on the dependent variable regardless of the specification used, while the unemployment rate shows the expected negative impact on growth. Openness to international trade is also shown not to be conducive to growth, which accords with a substantial segment of the economics literature that views the impact of trade with reservations.<sup>16</sup> A similar result is obtained for FDI, as it is shown by the pairing of the instrumental variables and GMM estimates with both the revenue and expenditure indicators of fiscal decentralization to have a strongly negative and significant impact on growth, underlining concerns about the impact of foreign investment in this respect. The impact of remittances is mostly positive and significant, as would be expected, but this variable may have the opposite impact when analysed from a dynamic perspective. Higher inflation, bigger government (as measured by the general government final consumption expenditures variable) and greater urbanization also seem to lead to less growth. Finally, greater political and civil liberties ratio) may have a more nuanced impact in a dynamic setting.<sup>17</sup>

Table 3 presents regression results showing how fiscal decentralization impacts growth when Canada and the United States are excluded from the analysis. The specifications and statistical properties of each regression are the same as in table 2.

As with the full sample of countries, fiscal decentralization on the revenue side is consistently shown to have a detrimental impact on growth. The coefficients for this variable are mostly negative, and they are statistically significant when estimated using GLS (with no effects and with random effects) and instrumental variables. On the expenditure side, the coefficients for the decentralization indicator are positive in all cases and are statistically significant when estimated using GLS (with fixed effects) and GMM, demonstrating that expenditure decentralization does seem to play an important role in generating economic growth in developing nations of the Americas. This result is more conclusive than the one observed with the full sample of countries, revealing that for this set of developing countries in the Americas, decentralization on the expenditure side may indeed have the expected positive impact on growth.

With respect to the control variables, for the most part their behaviour resembles the situation with the full sample of countries. Savings and the unemployment rate are consistently shown to have the expected positive and negative impacts on growth, respectively, while openness to trade, FDI and bigger government show a similar negative impact on growth. Remittances also seem to have a positive impact on growth, but, as with the full sample of countries, the impact of this variable in a dynamic setting might be more nuanced. Inflation is also shown to be (mostly) a negative influence on growth while, by contrast with the cases of Canada and the United States included in the analysis, the degree of urbanization seems to play no role in growth. Finally, greater political and civil liberties seem to foster economic growth, although, once again, the impact of this variable in a dynamic framework seems to be more nuanced.

Tables 4 and 5 analyse the impact of fiscal decentralization on the inflation rate. The regression methodologies are the same as those utilized when growth was the dependent variable. Table 4 presents estimation results for all the countries included in the analysis.

The impact of fiscal decentralization on growth, inflation and inequality in the Americas

<sup>&</sup>lt;sup>16</sup> A good survey of findings addressing some of the reservations about the impact of trade on growth is provided by Rodríguez and Rodrik (1999).

<sup>&</sup>lt;sup>17</sup> The ratio of political and civil liberties ranges from 0.18 (most free) to 1.00 (least free).

	Generalized least for cross-sectional heteroskedasticity (no effects s	squares to allow and intrasectional and autocorrelation pecification)	Generalized I with cross-se effects and	east squares ectional fixed no weights	Genera squares sectional ra and no	lized least with cross- andom effects weights <sup>b</sup>	Instrumental varia cross-sectional au heteroskedasticity a (no effects sp	tbles to allow for d intrasectional autocorrelation becification)	Generalize of mo	id method nents
	-0.074**		-0.065		-0.082**		-0.075*		0.006	
revenue aeceniralization (%)	(0.022)		(0.065)		(0.032)		(0.046)		(0.039)	
		0.026		0.183*		0.010		0.008		0.262***
EXperiorure decerritarization ( %)		(0.033)		(0.100)		(0.040)		(0.040)		(0.022)
	0.001*	-0.004**	0.001	-0.005**	0.001	-0.004**	-0.011	-0.002	-0.008	-0.005**
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)	(0.012)	(0.003)	(0000)	(0.001)
Proce domontic conjune (0/ of CDD)	0.191***	0.183**	0.209*	0.128	0.174***	0.126	0.154	0.181*	0.444***	0.269***
aioss autitestic savirigs (70 ur alit)	(0.057)	(0.080)	(0.118)	(0.191)	(0.061)	(0.093)	(0.107)	(0.100)	(0.082)	(0.060)
Occurrent to international trada (N + MM/CDD) (9/1	-0.033	-0.009	0.003	-0.036	-0.037*	0.002	-0.030	0.008	-0.225**	-0.101**
Uperifiess to international trade ( $(A + W)/ADP$ ) ( $\%$ )	(0.021)	(0.026)	(0:059)	(0.104)	(0.022)	(0:030)	(0.042)	(0.029)	(0.036)	(0.028)
Domittoncoc (0/ of CDD)	0.218***	-0.161	-0.062	-0.457	0.217*	-0.449	0.114	-0.190	0.173	-0.519**
	(0.078)	(0.244)	(0.408)	(0.547)	(0.124)	(0.404)	(0.145)	(0.343)	(0.228)	(0.200)
Ecroise direct investment $0/$ of $\Omega D$	0.056	-0.111	0.069	-0.514**	-0.001	-0.230	-0.305*	-0.364*	-0.062	-0.810**
	(0.076)	(0.102)	(0.162)	(0.247)	(0.130)	(0.185)	(0.171)	(0.192)	(0.084)	(0.051)
namalavimant rata /0/ af tatal labaur faraa)	-0.208**	-0.235**	-0.248*	-0.454**	-0.301**	-0.381**	-0.218**	-0.274**	-0.414**	-0.564**
	(0.061)	(0.067)	(0.133)	(0.157)	(0.102)	(0.102)	(0.103)	(0.124)	(0.085)	(0.031)
General government final consumption	0.067	-0.234**	-0.075	-0.561 **	0.072	I	0.011	-0.142	-1.254**	-0.636**
expenditure (% of GDP)	(0.073)	(0.082)	(0.247)	(0.226)	(0.121)	1	(0.146)	(0.144)	(0.185)	(0.049)
(letot) // antisticus (letotal)	-0.003	-0.030	-0.045	0.003		ı	0.005	-0.026	0.090	-0.001
	(0.037)	(0.039)	(0.166)	(0.399)		1	(0.062)	(0.053)	(0.094)	(0.108)
Dolition and aid libortion ratio	1.127	-5.116**	5.215	-2.949	1.639	-3.333	-0.869	-11.666*	8.810***	-6.516**
Political allo civil lidel les latio	(1.725)	(2.686)	(3.540)	(5.075)	(2.352)	(3.418)	(2.357)	(4.620)	(1.199)	(2.324)
Observations	163	123	163	123	163	130	144	114	134	105
Adjusted R <sup>2</sup>	0.27	0.26	0.09	0.21	0.11	0.15	0.17	0.23	ı	
Source: Prepared by the author on the basis o	of data from Interna (D) [online] www.c	ational Monetary Fu	ind (IMF) [onli lism/fiscal-de	ne] http://dat	a.imf.org/?s	k=E86E9088-3 +m Economic	1830-4CA3-B240-1	IBOEC5E15221; O atin Amarica and	rganization 1 the Caribh	or Economic

01:000 relization and GDP ner Table 3 hore form dare 01/00000

online] http://estadisticas.cepal.org/cepalstat/portada.html?idioma=english; Federal Tax Commission of Argentina [online] www.cfi.gov.ar; Ministry of Treasury and Public Finance of Argentina online] www.economia.gob.ar, National Institute of Statistics of the Plurinational State of Bolivia [online] www.ine.gob.bo; Freedom House [online] http://freedomhouse.org/report-types/ freedom-world#.VY\_fM1fRHcw; and World Bank [online] http://data.worldbank.org/data-catalog/world-development-indicators.

(\*) = significant at 10%, (\*\*) = significant at 5% and (\*\*\*) = significant at 1%. The dependent variable is per capita GDP growth. All regressions include an intercept term (not shown in table). Generalized method of moments specifications include a one-period lagged dependent variable as regressor (not shown in table). Standard errors are given in parentheses. Argentina, Brazil, Chile, Colombia, Costa Rica, El Salvador, Mexico, Paraguay, Peru and the Plurinational State of Bolivia. Note: g

Random effects estimation requires the number of cross-sections to be greater than the number of regressors. To fulfil this condition, the urban population and/or the general government final consumption expenditures variables were excluded from the specifications. p

	Generalized I. allow for cros intrasectional h and autocc effects sp	east squares to ss-sectional and leteroskedasticity mrelation (no ecification)	Generalized with cross-s effects anc	least squares tectional fixed I no weights	Generalized with cross-se effects and	l least squares sctional random d no weights	Instruments allow for cros intrasectional h and autoco effects sp	al variables to ss-sectional and eteroskedasticity prrelation (no ecification)	Generalized me	hod of moments
Dourse describeding (0/)	0.178**		6.102***		4.978***		-0.210**		-0.319***	
Hevenue decentralization (%)	(0.081)		(1.936)		(1.365)		(0.025)		(0.061)	
L'incordition d'accordination (00)		1.757***		-7.905		6.004**		1.788***		-8.589**
Experiorure decentralization (%)		(0.350)		(5.417)		(3.013)		(0.530)		(1.370)
GDP per capita (at purchasing	-5.363**	-36.269**	128.446**	24.671	-13.682	-87.547*	-2.063**	-30.795**	-7.795***	-75.204**
power parity, in logs)	(0.878)	(5.808)	(50.639)	(157.525)	(26.921)	(48.845)	(0.376)	(7.648)	(1.681)	(21.903)
Openness to international	-0.210**	0.344***	-6.754***	-2.697	-1.897**	0.549	-0.087**	0.443***	0.131 ***	1.045**
trade ((X + M)/GDP) (%)	(0.045)	(0.110)	(1.889)	(4.723)	(0.895)	(1.678)	(0.013)	(0.163)	(0.049)	(0.511)
General government	-2.710**	-7.516***	-72.671***	-61.180**	-47.331**	-36.530**	0.659***	-6.892***	0.368	-30.593**
tinal consumption expenditure (% of GDP)	(0.647)	(1.637)	(7.983)	(14.734)	(6.068)	(8.465)	(0.096)	(2.282)	(0.355)	(3.563)
	-2.018**	11.491***	103.508***	113.495	19.987	47.122*	-1.675**	11.178**	-2.147*	52.096***
INIIIIAI y experiurures (70 01 GDF)	(0.749)	(2.842)	(29.548)	(77.874)	(16.982)	(28.418)	(0.222)	(4.603)	(1.232)	(19.519)
Foreign direct investment	-0.660**	-5.327***	-7.410	3.698	-3.934	-12.981	-0.727**	-3.771***	-0.082	9.245***
(% of GDP)	(0.179)	(0.964)	(5.397)	(13.523)	(4.630)	(10.670)	(0.165)	(0.925)	(0.173)	(2.362)
	-0.880**	-10.006**	20.452*	39.263	-1.041	-20.477	-0.552**	-8.706***	-0.479	18.764***
Nellinualices (% 01 GDF)	(0.139)	(2.000)	(11.920)	(29.979)	(4.973)	(21.396)	(0.073)	(1.995)	(0.304)	(5.594)
Observations	178	149	178	149	178	149	159	139	148	129
Adjusted R <sup>2</sup>	0.20	0.13	0.39	0.23	0.23	0.12	0.57	0.03	1	I
Source: Prepared by the autho	or on the basis o	of data from Interné	ational Monetary	· Fund (IMF) [onlir aralism /fiscal_day	http://data.im/http://data.im	ft.org/?sk=E86E{ tabasa htm. Eco	3088-3830-4CA3- nomio Commissio	-B240-1B0EC5E1	15221; Organizat Vice and the Ca	ion for Economic
Cooperation and Dev	elopment (UEU	U) [online] www.c	secd.org/ctp/tea	eralism/tiscal-de	centralisation-da.	tabase.htm; Ecc	nomic Commissi	ION TOY LATIN AME	irica and the Ca	100ean (EULAU

**Table 4** The Americas (12 countries):<sup>a</sup> fiscal decentralization and inflation joninej nup.//estadisticas.cepa.org/cepastar/portada.nuni/rationna=english; redeta na Commission of Agenuna joninej www.cin.gov.ar; hinitistry of freadom and rupic frieadom and rupic freadom house.org/report-types/ [online] www.economia.gob.ar; National Institute of Statistics of the Plurinational State of Bolivia [online] www.ine.gob.bo; Freedom House [online] http://freedomhouse.org/report-types/ freedom-world#.VY\_MNTRHcw; and World Bank [online] http://data.worldbank.org/data-catalog/world-development-indicators. (\*) = significant at 10%, (\*\*) = significant at 5% and (\*\*\*) = significant at 1%. The dependent variable is the inflation rate. All regressions include an intercept term (not shown in table). Generalized

method of moments specifications include a one-period lagged dependent variable as regressor (not shown in table). Standard errors are given in parentheses. <sup>a</sup> Argentina, Brazil, Canada, Chile, Colombia, Costa Rica, El Salvador, Mexico, Paraguay, Peru, the Plurinational State of Bolivia and the United States. Note:

68 CEPAL Review Nº 124 • April 2018

The results for the effects of fiscal decentralization variables on the inflation rate are inconclusive. When decentralization occurs on the revenue side, the GLS estimates for the decentralization indicator are positive and statistically significant, implying that greater decentralization fosters higher inflation. However, when the specification is instrumented and the relationship is analysed within a dynamic setting, the impact of revenue decentralization is reversed, meaning that it actually has a dampening impact on inflation. With the expenditure decentralization indicator, the results are slightly clearer: GLS and instrumental variables estimates show a positive and statistically significant correlation between expenditure decentralization on inflation is reversed with GMM estimation, implying that the impact of expenditure decentralization on inflation is not entirely clear. The lack of clarity in the results precludes any definite conclusions as to how fiscal decentralization impacts inflation, and hence it cannot be unambiguously stated that it either deters or induces inflation in the countries of the Americas.

Regarding the control variables, per capita GDP growth is consistently associated with lower inflation, an unsurprising result in view of the findings of previous studies, such as Martínez-Vázquez and McNab (2006).<sup>18</sup> Government expenditures and FDI mostly act to quell inflation, while the results for the rest of the variables (openness to international trade, military expenditures and remittances) are inconclusive.

Table 5 presents results for the effects of fiscal decentralization on the inflation rate when Canada and the United States are excluded from the analysis.

As is evident, the results largely confirm the findings presented in table 4. Both fiscal decentralization indicators show the same pattern of behaviour as is observed with the full sample of countries, whence the difficulty of drawing any definite conclusions. Under certain conditions, as reflected in the positive and statistically significant GLS estimates, revenue decentralization seems to foster inflation, while in others, as seen when the specification is instrumented and when it is estimated with GMM, the opposite is true. Likewise, expenditure decentralization seems to be more conducive to higher inflation with GLS and instrumental variables estimates, but in a dynamic setting the opposite is true. As was concluded for the full sample of countries, it cannot be unequivocally stated that fiscal decentralization deters or induces inflation in developing nations of the Americas.

The behaviour of the control variables is slightly better defined with this sample of countries. As expected, GDP per capita is for the most part associated with lower inflation, as are government expenditures and FDI, demonstrating that the size of government and net capital inflows play a positive role in preventing inflation.<sup>19</sup> Military expenditures seem to be conducive to higher inflation, although when coupled with decentralization on the revenue side they may be a deterrent to it. Lastly, the impact of trade openness and remittances is indeterminate.

The last set of regressions is reported in tables 6 and 7. Table 6 presents results for the impact of fiscal decentralization on the Gini coefficient across the full sample of countries.

<sup>&</sup>lt;sup>18</sup> GLS with cross-sectional fixed effects reverses the overall trend and GDP per capita is shown to have a positive and significant impact on inflation. Since alternative GLS techniques, instrumental variables and GMM consistently generate negative and statistically significant coefficients for this variable, the result of GLS with fixed effects is taken to be an anomaly.

<sup>&</sup>lt;sup>19</sup> In a dynamic setting, FDI may be conducive to higher inflation, as evidenced in the GMM specification. Likewise, bigger government may also have a positive impact on inflation, as reflected in the instrumental variables estimate when decentralization occurs on the revenue side.

0.840***           Revenue decentralization (%)         0.301)           Expenditure decentralization (%)         0.301	skedasticity tion (no ation)	Generalized with cross-si effects and	east squares ectional fixed no weights	Generalized with cross-se effects and	least squares ctional random I no weights	allow for cross intrasectional hi and autoco effects sp	s-sectional and leteroskedasticity brrelation (no ecification)	Generalized me	thod of moments
revenue decentralization (%) (0.301) Expenditure decentralization (%)		6.706***		5.968***		-0.231**		-0.258***	
Expenditure decentralization (%)		(2.275)		(1.663)		(0.072)		(0.069)	
	1.816***		-7.900		3.343		3.575***		-6.783***
	(0.447)		(6.684)		(4.643)		(1.308)		(1.536)
GDP per capita <i>(at purchasing</i> 3.093 -1	15.348**	148.802**	38.037	45.479	66.310	-1.957*	12.006	-16.445**	-100.882**
power parity, in logs) (3.413)	(6.640)	(67.189)	(215.942)	(37.732)	(89.204)	(1.157)	(13.963)	(1.690)	(34.722)
Openness to international	0.028	-6.874***	-0.350	-3.000***	-2.273	-0.108**	0.599	0.192***	3.961***
trade $((X + M)/GDP)$ (%) (0.130)	(0.173)	(2.178)	(5.933)	(0.939)	(2.678)	(0.030)	(0.391)	(0.043)	(0.956)
General government final consumption -0.804** -1	11.084**	-75.553***	-57.450**	-56.733***	-55.139**	0.573*	-21.362**	0.330	-18.646***
expenditure (% of GDP) (2.085)	(3.136)	(9.086)	(17.974)	(7.563)	(11.727)	(0.303)	(7.886)	(0.387)	(4.124)
11.085** 3	37.026***	162.507***	272.964*	85.790***	186.603**	-2.463**	87.810**	-13.181**	234.145***
Minitary experioritories ( <i>76 of GDP</i> ) (5.421) (1	11.515)	(54.611)	(157.370)	(28.124)	(75.437)	(1.227)	(33.873)	(1.725)	(73.216)
-2.173**	-6.474***	-7.775	6.200	-9.297*	-21.243	-0.873**	-8.903**	-0.044	8.988***
	(1.468)	(6.471)	(16.999)	(5.585)	(13.312)	(0.308)	(3.433)	(0.171)	(2.305)
-0.332	-4.219*	23.112*	39.296	2.615	9.218	-0.588**	-9.347*	-1.151***	17.554**
Neminuarices ( 20 0/ GUP) (0.386)	(2.445)	(13.477)	(34.567)	(4.768)	(28.332)	(0.104)	(5.125)	(0.304)	(7.877)
Observations 141 11	14	141	114	141	114	124	106	115	98
Adjusted R <sup>2</sup> 0.17	0.07	0.40	0.23	0.28	0.19	0.25	I	I	I

ionline] www.economia.gob.ar; National Institute of Statistics of the Purinational State of Bolivia [online] www.ine.gob.bo; Freedom House [online] http://freedomhouse.org/report-types/

freedom-world#.VY\_fM11RHcw; and World Bank [online] http://data.worldbank.org/data-catalog/world-development-indicators.

Note: (\*) = significant at 10%, (\*\*) = significant at 5% and (\*\*\*) = significant at 1%. The dependent variable is the inflation rate. All regressions include an intercept term (not shown in table). Generalized method of moments specifications include a one-period lagged dependent variable as regressor (not shown in table). Standard errors are given in parentheses.
<sup>a</sup> Argentina, Brazil, Chile, Colombia, Costa Rica, El Salvador, Mexico, Paraguay, Peru and the Plurinational State of Bolivia.

Table 5 The Americas (10 countries):<sup>a</sup> fiscal decentralization and inflation

The impact of fiscal decentralization on growth, inflation and inequality in the Americas

70 CEPAL Review N° 124 • April 2018

		TheA	mencas (12	countries):4	nscal decen	tralization a	and income	distribution			
		Generalized le allow for cross intrasectional he and autoc (no effects s	ast squares to -sectional and teroskedasticity orrelation pecification)	Generalized with cross-si effects and	least squares ectional fixed no weights	Generalized I with cross-sec effects and	east squares :tional random no weights <sup>b</sup>	Instrumenta allow for cross intrasectional he and autocol effects sp	variables to sectional and teroskedasticity relation (no scification)	Generalized me	hod of moments
	100	-0.049**		0.012		-0.058**		-0.027		-0.022	
Hevenue	uecentralization (%)	(0.014)		(0.053)		(0.022)		(0.029)		(0.049)	
- Hiboood	(//)		-0.004		0.108		-0.016		-0.007		0.008
Expendim	e decentralization (%)		(0.027)		(0.068)		(0.028)		(0.052)		(0.040)
GDP per c.	apita (at purchasing	-1.684	27.187*	-102.864**	-90.425**	5.712	36.777***	-0.489	13.469	-26.533*	-40.762*
power par	ity, in logs)	(6.841)	(14.731)	(23.568)	(28.058)	(10.665)	(12.641)	(11.766)	(30.313)	(15.422)	(19.504)
GDP per c.	apita squared <i>(at</i>	-0.319	-1.783**	5.051 ***	4.449***	-0.704	-2.300***	-0.358	-1.011	1.366*	2.049**
purchasin	g power parity, in logs)	(0.369)	(0.793)	(1.247)	(1.493)	(0.572)	(0.673)	(0.641)	(1.635)	(0.801)	(0.960)
	10/ of total	0.232***	-0.084	0.452**	0.995*	0.164***	-0.039	0.444***	-0.040	0.099	-0.055
UIDAII POL	- 	(0.036)	(0.085)	(0.218)	(0.539)	(0.046)	(0.063)	(0.086)	(0.153)	(0.107)	(0.347)
Openness	to international	0.023*	-0.010	0.024	-0.030		1	0.095***	0.017	0.053**	0.105**
trade ((X +	- M)/GDP) <i>(%)</i>	(0.014)	(0.029)	(0.049)	(0.081)			(0.035)	(0.042)	(0.022)	(0.052)
Dolition or	the state of the second s	3.332**	-0.917	-12.170***	-14.835**	0.191	-3.234	9.609***	-2.484	1.866	4.292
Puilleal a		(1.536)	(2.233)	(3.848)	(3.882)	(2.329)	(2.699)	(3.251)	(6.114)	(2.160)	(2.667)
Domittono		-0.403**	0.306	-0.111	-0.423	-0.335**	0.202	-0.804***	0.198	-0.470**	-0.948**
חפוווונמוור		(0.074)	(0.194)	(0.310)	(0.408)	(0.100)	(0.260)	(0.116)	(0.295)	(0.195)	(0.235)
Groce dow	instic conjuga (0/ of CDD)	-0.131**	-0.338***	0.109	0.103	-0.046	-0.333***	-0.380***	-0.431**	-0.108*	0.115
		(0.045)	(0.085)	(0.105)	(0.156)	(0.053)	(0.084)	(0.095)	(0.157)	(0.061)	(0.085)
General gu	vernment final consumption	0.422***	0.074	0.168	0.148	0.429***	0.214**	0.504***	0.209	0.076	0.302*
expenditur	e (% of GDP)	(0.067)	(0.089)	(0.248)	(0.220)	(0.120)	(0.111)	(0.124)	(0.188)	(0.184)	(0.182)
Inflation ra	( /0/ ·+	-0.077**	0.047	-0.080	0.023	-0.133**	ī	-0.337***	0.236	-0.070**	-0.147**
	- <i>(%)</i>	(0.024)	(0:030)	(0:050)	(0.051)	(0.038)	1	(20.097)	(0.171)	(0.018)	(0.033)
Intornat up	(classes 100 root)	-0.012	0.049*	-0.041	-0.045	-0.027	0.029	-0.083***	0.016	-0.037*	-0.020
ווונפו וופר חי	sers (her ino heapie)	(0.016)	(0.027)	(0.042)	(0.055)	(0.019)	(0.022)	(0.027)	(0.052)	(0.021)	(0.029)
molocion	mont rate (0% of total labour forma)	0.155***	0.164**	0.124	0.439**	0.231**	0.098	0.123	0.063	-0.046	0.143*
uireiripiny.	IIIEIILIAIE (70 01 101a1 101001 10106)	(0.058)	(0.079)	(0.210)	(0.190)	(0.105)	(0.095)	(0.150)	(0.183)	(0.144)	(0.086)
Observatic	SUC	171	139	171	139	171	139	154	128	142	117
Adjusted F	22	0.81	0.81	0.84	0.81	0.61	0.61	0.78	0.85	ı	I
Source:	Prenared by the author on the F	asis of data fro	m International 1	Monetary Fund (	(IMF) [online] httr	v://data imf ord/	7.sk=F86F9088	-3830-4CA3-B2	40-1 B0FC5F15	221: Organizati	on for Economic
	Cooperation and Development	(OECD) [online	] www.oecd.or	g/ctp/federalism	Vfiscal-decentral	isation-databas	e.htm; Econom	ic Commission	for Latin Americ	a and the Car	bbean (ECLAC)
	online] http://estadisticas.cepa	.org/cepalstat/p	bortada.html?idi	oma=english; Fe	ederal Tax Comn	nission of Argen	tina [online] ww	w.cfi.gov.ar; Mini	stry of Treasury	and Public Fina	nce of Argentina
+-	reedom-world# VY fW11BHcw	- and World Bar	ald of Oliariasius ak fonlinel http:/	/data worldhank	c ora/data-catalc	arivia (orimio) wy arivorid-develoi	oment-indicator	2 2			Bricholt types
Note:	*) = significant at 10%, (**) = sig The political and civil liberties rat	in range from	and $(***) = signification = $	cant at 1%. The	dependent varia	able is the Gini o	coefficient, which	ranges from 0	(complete equal	tty) to 100 (com	plete inequality).
				וח ויחח וובמאו ווב	101000 Lenies	יו ווחומ שווחות	בורפחר ופוווו (ווחר	SILOWIT III LAUIEJ.	מפו ופו מוודפת וו ופ		rs shealladin is

5 ÷ • ÷ Table 6

Note:

include a one-period lagged dependent variable as regressor (not shown in table). Standard errors are given in parentheses. Argentina, Brazil, Canada, Chile, Colombia, Costa Rica, El Salvador, Mexico, Paraguay, Peru, the Plurinational State of Bolivia and the United States. Random effects estimation requires the number of cross-sections to be greater than the number of regressors. To fulfil this condition, the openness to international trade and/or inflation rate variables

were excluded from these specifications.

p a

The impact of fiscal decentralization on the Gini coefficient varies depending on whether decentralization occurs on the revenue or expenditure side. There is substantial evidence that decentralization on the revenue side plays a positive role in creating conditions for greater equality, as evinced by the negative, statistically significant and approximately same-sized coefficients obtained for the revenue decentralization indicator using GLS specifications with no effects and with random effects. The impact of expenditure decentralization, on the other hand, seems to be non-existent, as the coefficients estimated for this variable are in all cases close to zero and statistically insignificant. The general conclusion, therefore, is that revenue decentralization seems to play the expected positive role in generating conditions for greater income equality, whereas decentralization on the expenditure side seems to be ineffective in addressing inequality concerns.

The behaviour of the control variables is also noteworthy. GDP per capita in levels and GDP per capita squared do not seem to follow a clear pattern of behaviour, and hence it is not possible to reach any definite conclusions about the way these variables impact income inequality. The uniform behaviour pattern of the GDP variable suggests, however, that the Kuznets hypothesis, implying greater inequality in early stages of development but less inequality in more advanced stages, may not apply here. There is convincing evidence that greater urbanization, openness to international trade, larger government and higher unemployment worsen income inequality, as reflected in the consistently positive and statistically significant coefficients for these variables. These results fall neatly into line with findings elsewhere (see, for instance, Rodríguez and Rodrik, 1999; Lee, 2005; Martínez, Ayala and Ruiz-Huerta, 2001) that urbanization, open markets, an increasingly active government and higher unemployment are the principal drivers of greater income inequality. Savings and remittances, on the other hand, seem to be conducive to greater income equality, reflected in consistently negative and statistically significant coefficients, a result that is equally unsurprising given the importance of deferred consumption and alternative sources of income as instruments for repressing income inequality. Inflation also seems to assuage inequality, as reflected in negative and statistically significant estimates for this variable with different techniques. This result is consistent with findings elsewhere (e.g. Monnin, 2014) that monetary policies aimed at controlling inflation have operated to the detriment of people in the middle and lower income brackets and have therefore increased income inequality. The argument is that whenever wages accelerate and central banks tighten monetary policy, unemployment rises, implying a worsening of income inequality. The counter-argument, then, is that higher inflation may reduce income inequality by allowing higher wage growth. Lastly, the political and civil liberties and Internet users ratios do not follow any discernible pattern, so that it is not possible to draw any conclusions about their impact on income inequality.

Table 7 reports regression estimates for the impact of fiscal decentralization on the Gini coefficient in countries of the Americas, excluding Canada and the United States.

The impact of fiscal decentralization on the Gini coefficient is clearer with this subset of countries in the Americas. Both decentralization indicators are consistently positive, of similar size and, in several cases, statistically significant, implying that decentralization may actually contribute to greater inequality in developing nations of the Americas. Although this implication seems to be stronger when decentralization occurs on the expenditure side, the general conclusion in both cases seems to be that fiscal decentralization has not delivered on its promise of contributing to greater income equality in these countries, and this, as Brosio and Jiménez (2013) rightly point out, highlights the need to strengthen coordination mechanisms and arrangements between all levels of government to ensure more efficiency and better delivery of outcomes in the decentralization structures of these countries.

Revenue decentralization (%) $0.037^{++}$ $0.006$ Expenditure decentralization (%) $0.017$ ) $0.081^{+++}$ $0.058$ )           Expenditure decentralization (%) $0.031^{++}$ $0.058^{++}$ $0.0530^{++}$ GDP per capita (at purchasing $0.030^{+}$ $-29.114^{+}$ $1.436^{+}$ power parity, in logs) $(9.432)$ $(19.170)$ $(35.551)^{+}$ GDP per capita squared ( $5.059^{+++}$ $4.954^{+++}$ $1.436^{-}$ at purchasing power parity, in logs) $(0.531)$ $(1.072)$ $(1.903)^{-}$ Urban population (% of total) $(0.059)$ $(0.085)^{-}$ $0.223^{-}$ Urban population (% of total) $(0.059)^{-}$ $(0.077)^{-}$ $0.226^{++++}$ $1.436^{-}$ Optical and civil liberties ratio $(0.059)^{-}$ $(0.085)^{-}$ $(0.133)^{+}$ $(0.028)^{-}$ $(0.132)^{+}$ Political and civil liberties ratio $(1.712)^{-}$ $(2.189)^{-}$ $(1.33)^{+}$ $(1.73)^{-}$ Remittances (% of GDP) $(0.019)^{-}$ $(0.028)^{-}$ $(0.026)^{-}$ $(0.026)^{-}$ Remittances (% of GDP)	0.006 0.058) 0.058) 0.048 0.090) 9.114 -57.949 6.551) (57.967) 1.436 2.986 1.903) 0.223 0.23 0.246) 0.269 0.036 0.056 0.030 0.036 0.030 0.036 0.030 0.036 0.0300 0.0300 0.0300 0.030000 0.0300000000	0.030	o weights <sup>u</sup>	heteroskedasticity (no effects s	and autocorrelation specification)		d method nents
$(0.017)$ $(0.058)$ Expenditure decentralization (%) $(0.051)$ $(0.053)$ GDP per capita (at purchasing power parity, in logs) $(9.492)$ $(19.170)$ $(36.551)$ Dower parity, in logs) $(9.492)$ $(19.170)$ $(36.551)$ Dower parity, in logs) $(9.492)$ $(19.170)$ $(36.551)$ Date capita squared ( $\chi$ $(0.059)$ $(0.072)$ $(1.303)$ Urban population (% of total) $(0.059)$ $(0.085)$ $(0.246)$ Openess to international $0.003$ $-0.042^*$ $-0.013$ Dentess to international $0.003$ $-0.048^*$ $-0.013$ Dentess to international $0.003$ $-0.048^*$ $-0.013$ Dentess to international $0.003$ $-0.048^*$ $-0.013$ Remittances (% of GDP) $(0.019)$ $(0.028)$ $(0.172)$ Remittances (% of GDP) $0.0022$ $-0.013$ $(0.033)$ Remittances (% of GDP) $0.0022$ $-0.022$ $-0.013$ Gases domestic savings (% of GDP) $0.0022$ $-0.022$ $-0$	0.058) 0.058) 0.048 0.090) 0.114 -57.949 6.551) 1.436 2.986 1.903) 0.226) 0.026 0.036 0.036 0.036 0.036 0.036 0.036 0.055 0.036 0.055 0.036 0.056 0.056 0.056 0.036 0.056 0.	(0.021)		0.056		-0.005	
Expenditure decentralization (%) $0.081^{***}$ Expenditure decentralization (%) $0.030$ )           GDP per capita (at purchasing power parity, in logs) $90.537^{*}$ $88.800^{**}$ $-29.114$ GDP per capita (at purchasing power parity, in logs) $9.492$ ) $(19.170)$ $(36.551)$ GDP per capita squared ( $5.059^{***}$ $4.954^{***}$ $1.436$ GDP per capita squared ( $5.059^{***}$ $4.954^{***}$ $1.436$ Urban population (% of total) $(0.531)$ $(1.072)$ $(1.322)$ Urban population (% of total) $(0.035)$ $(0.013)$ $0.223$ Urban population (% of total) $(0.033)$ $(0.026)$ $(0.133)$ Demess to international $0.003$ $-0.048^{*}$ $-0.013$ Urban population (% of GDP) $(0.023)$ $(0.028)$ $(0.055)$ Political and civil liberties ratio $(1.712)$ $(2.189)$ $(0.055)$ Political and civil liberties ratio $(1.712)$ $(2.029)$ $(0.138)$ Gorss domestic savings (% of GDP) $(0.022)$ $(0.026)$ $(0.026)$ G	0.048 0.090) 0.114 -57.949 0.090 0.090) 0.090) 0.090 0.091 0.003 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.055 0.003 0.005	(1 30.0)		(0.042)		(0.059)	
(0.030)         (0.030)           GDP per capita (at purchasing         -90.537*         -88.800**         -29.114           power parity, in logs)         (9.492)         (19.170)         (36.551)           GDP per capita squared (         5.059***         4.954***         1.436           GDP per capita squared (         5.059***         4.954***         1.436           GDP per capita squared (         5.059***         4.954***         1.436           Urban population (% of total)         (0.059)         (0.085)         (0.246)           Openness to international         0.003         -0.048*         -0.013           Urban population (% of total)         0.003         -0.048*         -0.013           Openness to international         0.003         -0.048*         -0.013           Itade (X + M//GDP) (%)         (0.019)         (0.028)         (0.055)           Political and civil liberties ratio         1.1712         (2.189)         (4.133)           Remittances (% of GDP)         0.174**         0.246         0.036           Gross domestic savings (% of GDP)         0.174**         0.2602         0.0403           Gross domestic savings (% of GDP)         0.174**         0.267         0.2607           Gross domestic savings	(0.090)           9.114         -57.949           6.551)         (57.949)           1.436         2.986           1.433         0.029)           0.246)         0.780           0.233         0.780           0.246)         0.659           0.013         -0.036           3.182**         -15.605**		0.131 ***		0.170***		-0.051
GDP per capita (at purchasing power parity, in logs)         -90.537*         -88.800**         -29.114           power parity, in logs) $(9.492)$ $(19.170)$ $(36.551)$ GDP per capita squared (at purchasing power parity, in logs) $(9.492)$ $(19.170)$ $(36.551)$ GDP per capita squared (at purchasing power parity, in logs) $(0.531)$ $(1.072)$ $(1.436)$ Urban population (% of total) $(0.059)$ $(0.085)$ $(0.246)$ Urban population (% of total) $(0.019)$ $(0.028)$ $(0.055)$ Urban population (% of total) $(0.019)$ $(0.028)$ $(0.055)$ Denness to interrational $0.003$ $-0.048^*$ $-0.013$ Urban polation (% of GDP) $(0.019)$ $(0.028)$ $(0.055)$ Political and civil liberties ratio $(1.712)$ $(2.189)$ $(4.133)$ Remittances (% of GDP) $(0.021)$ $(0.022)$ $(0.036)$ Remittances (% of GDP) $(0.021)$ $(0.242)$ $(0.026)$ Gross domestic savings (% of GDP) $(0.022)$ $(0.022)$ $(0.022)$ Gross domestic savings (% of GDP)	9.114         -57.949           6.551         (57.967)           1.436         2.986           1.436         2.986           0.223         0.780           0.2246         (0.669)           0.013         -0.036           0.055         (0.093)           3.182**         -16.065**		(0.034)		(0.056)		(0.032)
GDP per capits squared ( $(9.492)$ $(19.170)$ $(36.551)$ GDP per capits squared ( $5.059^{***}$ $4.954^{***}$ $1.436$ at purchashing power parity, in logs) $(0.531)$ $(1.072)$ $(1.903)$ Urban population (% of total) $(0.531)$ $(1.072)$ $(1.903)$ Urban population (% of total) $(0.55)$ $(0.246)$ $0.223$ Openness to international $(0.035)$ $(0.286)$ $(0.246)$ Openness to international $0.003$ $-0.048^{*}$ $-0.013$ Political and civil liberties ratio $5.682^{**}$ $8.775^{**}$ $-13.183^{**}$ Political and civil liberties ratio $(1.712)$ $(0.028)$ $(0.132)$ Remittances (% of GDP) $(0.019)$ $(0.021)$ $(0.342)$ Gross domestic savings (% of GDP) $0.174^{***}$ $0.210^{**}$ $0.342)$ Gross domestic savings (% of GDP) $(0.022)$ $(0.033)$ $(0.1112)$ Gross domestic savings (% of GDP) $(0.022)$ $(0.023)$ $(0.023)$ Gross domestic savinge (% of GDP) $(0.022)$ $(0$	6.551         (57.967)           1.436         2.986           1.903         0.780           0.223         0.780           0.246         (0.669)           0.013         -0.036           0.055         (0.093)           3.182**         -1, 606**	-88.140**	-93.186**	-53.609**	-139.377**	1.465	-30.495*
GDP per capita squared ( at purchashing power partly, in logs) $0.531$ $4.954^{***}$ $1.436$ at purchashing power partly, in logs) $(0.531)$ $(1.072)$ $(1.903)$ Urban population (% of total) $0.077$ $-0.426^{***}$ $0.226$ Urban population (% of total) $(0.059)$ $(0.085)$ $(0.246)$ Openness to international $0.003$ $-0.048^{*}$ $-0.013$ Depenness to international $0.003$ $-0.048^{*}$ $-0.013$ Political and civil liberties ratio $5.682^{**}$ $8.775^{**}$ $-13.183^{**}$ Political and civil liberties ratio $0.003$ $0.048^{*}$ $-0.013$ Remittances (% of GDP) $(0.081)$ $(0.201)$ $(0.342)$ Gross domestic savings (% of GDP) $0.174^{***}$ $0.210^{**}$ $0.342$ Gross domestic savings (% of GDP) $(0.022)$ $(0.033)$ $(0.112)$ Gross domestic savings (% of GDP) $0.174^{***}$ $0.262^{**}$ $(0.033)$ Gross domestic savings (% of GDP) $(0.022)$ $(0.023)$ $(0.023)$ Gross domestic savings (% of GDP)	1.436         2.986           1.903)         (3.029)           0.223         0.780           0.246)         (0.669)           0.013         -0.036           0.055)         (0.093)           3.182**         -1.606**	(19.528)	(36.547)	(17.102)	(34.705)	(24.175)	(18.857)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1.903) (3.029) 0.223 0.780 0.246) (0.569) 0.013 -0.036 0.055) (0.093) 3.182** -15.065**	4.874***	5.175**	2.922***	7.869***	0.165	1.789*
Urban population (% of total) $-0.077$ $-0.426^{***}$ $0.223$ Openness to international $(0.059)$ $(0.085)$ $(0.246)$ Openness to international $0.003$ $-0.048^*$ $-0.013$ Political and civil liberties ratio $-5.682^{***}$ $8.775^{***}$ $-13.183^{***}$ Political and civil liberties ratio $-5.682^{***}$ $8.775^{***}$ $-13.183^{***}$ Remittances (% of GDP) $(0.172)$ $(2.189)$ $(4.133)$ Remittances (% of GDP) $(0.172)$ $(2.189)$ $(4.133)$ Gross domestic savings (% of GDP) $0.114^{***}$ $0.210^{**}$ $(0.342)$ Gross domestic savings (% of GDP) $0.114^{***}$ $0.210^{**}$ $0.220^{**}$ Gross domestic savings (% of GDP) $0.124^{***}$ $0.367^{***}$ $0.220^{**}$ Gross domestic savings (% of GDP) $0.0022$ $0.0023$ $0.033^{***}$ $0.271^{**}$ Inflation rate (%) $0.028^{**}$ $0.028^{**}$ $0.076^{**}$ $0.076^{**}$	0.223 0.780 0.246) (0.669) 0.013 -0.036 0.055) (0.093) 3.182** -15.065**	(1.106)	(2.063)	(0.968)	(1.904)	(1.263)	(0.967)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	0.246) (0.669) 0.013 -0.036 0.055) (0.093) 3.183** -15.995**	-0.073*	-0.398***	0.114	-0.637***	-0.068	0.395*
$ \begin{array}{c cccc} \mbox{Openness to international} & 0.003 & -0.048^{*} & -0.013 \\ \mbox{trade} ((\chi + M)/GDP) (\%) & (0.019) & (0.028) & (0.055) \\ \mbox{Political and civil liberties ratio } & -5.682^{**} & -8.775^{***} & -13.183^{**} \\ \mbox{Political and civil liberties ratio } & -5.682^{**} & -8.775^{***} & -13.183^{**} \\ \mbox{Remittances } (\% of GDP) & (1.712) & (2.189) & (4.133) \\ \mbox{Remittances } (\% of GDP) & 0.057 & -0.166 & -0.403 \\ \mbox{Gons domestic savings } (\% of GDP) & 0.174^{***} & 0.210^{**} & 0.090 \\ \mbox{Gons domestic savings } (\% of GDP) & 0.174^{***} & 0.210^{**} & 0.000 \\ \mbox{Gons domestic savings } (\% of GDP) & 0.174^{***} & 0.210^{**} & 0.020 \\ \mbox{General government final consumption } & 0.589^{***} & 0.367^{***} & 0.262 \\ \mbox{Gons diffution rate } (\%) & (0.093) & (0.023) & (0.053) \\ \mbox{Inflation rate } (\%) & (0.023) & (0.023) & (0.053) \\ \end{tabular}$	0.013 -0.036 0.055 (0.093) 3.183** -15.995**	(0.043)	(0.068)	(0.116)	(0.152)	(0.102)	(0.214)
Political and civil liberties ratio         (0.019)         (0.028)         (0.055)           Political and civil liberties ratio         -5.682**         -8.775***         -13.183**           Remittances (% of GDP)         (1.712)         (2.189)         (4.133)           Remittances (% of GDP)         (0.057)         -0.166         -0.403           Gross domestic savings (% of GDP)         (0.114***         0.210**         0.990           General government final consumption         0.589***         0.367***         0.262           Renet government final consumption         0.589***         0.367***         0.262           Inflation rate (%)         (0.064)         (0.063)         (0.271)	0.055) (0.093) 3.183** -15.995**		ı	0.048	-0.022	0.044	-0.037
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	3 183** -15 995**	1		(0.052)	(0.052)	(0.037)	(0.029)
Remittances (% of GDP)         (1.712)         (2.189)         (4.133)           Remittances (% of GDP) $-0.057$ $-0.166$ $-0.403$ Gross domestic savings (% of GDP) $(1.712)$ $(2.201)$ $(0.342)$ Gross domestic savings (% of GDP) $0.114^{+++}$ $0.210^{++}$ $0.090$ General government final consumption $0.589^{+++}$ $0.367^{+++}$ $0.262$ expenditure (% of GDP) $0.589^{+++}$ $0.367^{+++}$ $0.262$ Inflation rate (%) $-0.127^{++}$ $-0.022$ $-0.76$ $(0.023)$ $(0.023)$ $(0.023)$ $(0.053)$		-9.956***	-12.356**	-2.497	-8.456*	-5.104*	-3.133*
Remittances (% of GDP) $-0.057$ $-0.166$ $-0.403$ Gross domestic savings (% of GDP) $(0.081)$ $(0.201)$ $(0.342)$ Gross domestic savings (% of GDP) $0.174^{***}$ $0.210^{**}$ $0.090$ Gross domestic savings (% of GDP) $0.174^{***}$ $0.210^{**}$ $0.090$ General government final consumption $0.589^{***}$ $0.367^{***}$ $0.262$ expenditure (% of GDP) $0.064$ $(0.093)$ $(0.271)$ Inflation rate (%) $-0.127^{**}$ $-0.022$ $-0.076$	4.133) (4.472)	(2.407)	(3.225)	(2.535)	(4.873)	(2.940)	(1.758)
(0.081)         (0.201)         (0.342)           Gross domestic savings (% of GDP)         0.174***         0.210**         0.090           General government final consumption         0.002)         (0.086)         (0.112)           General government final consumption         0.589**         0.367**         0.262           expenditure (% of GDP)         (0.064)         (0.093)         (0.271)           Inflation rate (%)         -0.127**         -0.022         -0.076	0.403 -0.523	T	1	-0.303**	-0.591	-0.723**	-0.698**
Gross domestic savings (% of GDP)         0.174***         0.210**         0.090           Gross domestic savings (% of GDP)         (0.002)         (0.086)         (0.112)           General government final consumption         0.589**         0.367***         0.262           expenditure (% of GDP)         (0.064)         (0.093)         (0.271)           Inflation rate (%)         -0.122**         -0.022         -0.076           (0.027)         (0.028)         (0.053)         (0.053)	0.342) (0.479)		1	(0.127)	(0.447)	(0.244)	(0.109)
(0.002)         (0.086)         (0.112)           General government final consumption         0.589***         0.367***         0.262           expenditure (% of GDP)         (0.064)         (0.093)         (0.271)           Inflation rate (%)         -0.127**         -0.022         -0.076           (0.027)         (0.028)         (0.053)         (0.053)	0.090 0.073	0.226***	060.0	0.044	0.557***	-0.126**	0.061
General government final consumption         0.589***         0.367***         0.262           expenditure (% of GDP)         (0.064)         (0.093)         (0.271)           Inflation rate (%)         -0.127**         -0.022         -0.076           (0.027)         (0.028)         (0.053)         (0.053)	0.112) (0.178)	(0.039)	(0.107)	(0.122)	(0.195)	(0.057)	(0.053)
(0.064)         (0.093)         (0.271)           Inflation rate (%)         -0.127**         -0.022         -0.076           (0.027)         (0.028)         (0.053)	0.262 0.209	0.549***	1	0.569***	0.626***	-0.051	0.015
Inflation rate (%)         -0.127**         -0.022         -0.076           (0.027)         (0.028)         (0.053)         (0.053)	0.271) (0.249)	(0.108)		(0.119)	(0.191)	(0.237)	(0.096)
(0.027) (0.028) (0.053)	0.076 0.052	1	I	-0.402***	-0.120	-0.059**	0.013
	0.053) (0.058)	ı		(0.087)	(0.127)	(0.026)	(0.017)
Internet users (per 100 people) -0.193** -0.086*** -0.131**	0.131** -0.112	-0.193***	-0.052	-0.214***	-0.129***	-0.135**	-0.105**
(0.011) (0.028) (0.055)	0.055) (0.080)	(0.025)	(0.037)	(0.023)	(0.045)	(0.029)	(0.019)
Unemployment rate (% of total labour force) 0.416*** 0.491*** 0.298	0.298 0.601 **	0.507***	0.630***	0.458**	0.917***	0.739***	0.404***
(0.083) (0.083) (0.249)	0.249) (0.229)	(0.106)	(0.119)	(0.186)	(0.193)	(0.119)	(0.065)
Observations 141 109 141	1 109	149	109	126	100	116	91
Adjusted R <sup>2</sup> 0.76 0.67	0.67 0.55	0.50	0.39	0.79	0.78		ı
Source: Prepared by the author on the basis of data from International Monetary Fund Cooperation and Development (OECD) [online] www.oecd.org/ctp/federalism	ry Fund (IMF) [online] http ederalism/fiscal-decentrali	://data.imf.org/? isation-database.	sk=E86E9088- .htm; Economi	-3830-4CA3-B240	D-1B0EC5E15221	; Organization and the Caribb	for Economic ean (ECLAC)

**Table 7** (10 countries):<sup>a</sup> fiscal decentralization and income

Note:

<sup>b</sup> Argentina, Brazil, Chile, Colombia, Costa Rica, El Salvador, Mexico, Paraguay, Peru and the Plurinational State of Bolivía.
 <sup>b</sup> Random effects estimation requires the number of cross-sections to be greater than the number of regressors. To fulfil this condition, the international trade openness, remittances, inflation and, for

expenditure decentralization, general government final consumption expenditure were excluded from these specifications.

(\*) = significant at 10%, (\*\*) = significant at 5% and (\*\*\*) = significant at 1%. The dependent variable is the Gini coefficient, which ranges from 0 (complete equality) to 100 (complete inequality). The political and civil liberties ratio ranges from 0.18 (most free) to 1.00 (least free). All regressions include an intercept term (not shown in table). Generalized method of moments specifications

With respect to the behaviour of the control variables, GDP per capita seems to contribute initially to greater income equality and eventually to a yet further increase, reflected in consistently positive and statistically significant coefficients for GDP per capita squared, a development that clearly goes against Kuznets's prediction. Greater urbanization, remittances and Internet access are all important factors contributing to greater income equality, as reflected in consistently negative and statistically significant coefficients, and this highlights the importance of urban settings, diaspora involvement and access to the World Wide Web as key determinants in lowering income inequality in developing countries. Larger government and higher unemployment, on the other hand, seem to be conducive to greater income inequality, as reflected in consistently positive and statistically significant coefficients, demonstrating that the quality rather than the size of government determines its productivity and highlighting the wellresearched fact that unemployment always tends to make matters worse. As with the full sample of countries, and for similar reasons, inflation also seems to have a dampening impact on inequality, while savings may contribute to greater inequality, perhaps because in developing nations of the Americas only a minority are able to save while the majority are simply unable to postpone current consumption. A curious result is the apparent effect of political and civil liberties on income inequality in these countries. The consistently negative and statistically significant coefficients for the political and civil liberties ratio seem to imply that the fewer political and civil liberties there are, the greater the equality of income. This result may reflect the perception in many developing countries of the Americas that strong governments are often needed to pass necessary legislation that may be unpopular at first but that will in time bring benefits, such as greater income equality. Finally, there is some indication that economic openness assuages income inequality.

### VI. Conclusions and policy implications

This article analyses the impact of fiscal decentralization on economic growth, inflation and income inequality in a sample of countries in the Americas. Given that most of these countries have undergone a period of gradual decentralization of economic functions to regional levels of government over past years, the results presented here seem timely and relevant.

The main findings are as follows. With respect to the impact of fiscal decentralization on economic growth when all countries are included in the analysis, the revenue-based decentralization indicator consistently shows a negative impact on growth. On the expenditure side, the evidence is inconclusive, and hence it cannot be determined whether decentralization has had a positive or negative impact on growth. When the same analysis is done without Canada and the United States, decentralization on the revenue side is consistently shown to exert a negative influence on growth, confirming the results obtained with the full sample of countries, but the expenditure decentralization indicator seems to show a positive effect on growth, demonstrating that decentralization on the expenditure side is similar regardless of the sample of countries being analysed. National saving, for instance, is consistently shown to contribute to growth, while higher unemployment, bigger government and greater openness to trade cause it to diminish. The impact of the remaining control variables is not as clear, and hence their combined effect on growth is indeterminate.

With respect to the impact of fiscal decentralization on inflation, the results are indeterminate, regardless of the sample of countries being analysed and of whether decentralization occurs on the revenue or expenditure side. In all cases, there is no discernible pattern of behaviour allowing unequivocal conclusions to be drawn about the ways in which decentralization impacts price stability. The behaviour of the control variables in both sets of regressions is similar, with the impact of these variables being slightly better defined in the sample of countries that excludes Canada and the United States. A noteworthy

result is that while government expenditures in both samples of countries generally seem to have a dampening influence on inflation, military expenditures seem to be more inflationary in the sample of developing countries of the Americas, highlighting the need to manage this type of expenditure with care. The impact of the remaining control variables is inconclusive.

Regarding the impact of fiscal decentralization on the Gini coefficient, finally, and considering the full sample of countries, the results indicate that decentralization on the revenue side has the expected positive impact in reducing income inequality, as reflected in consistently negative and statistically significant coefficients for GLS estimates (GLS specifications estimated with no effects and with random effects) that point to the importance of fiscal decentralization and its potential benefits in reducing income disparities. On the expenditure side, however, there is no evidence that decentralization has played any role in mitigating income inequalities.

The impact of fiscal decentralization on the dependent variable is somewhat clearer when Canada and the United States are excluded, regardless of whether decentralization occurs on the revenue or expenditure side. Fiscal decentralization indicators are generally positive and in some cases statistically significant (GLS specifications with no effects and with random effects and instrumental variables with no effects), demonstrating that fiscal decentralization has not played its expected role in reducing income inequality in developing nations of the Americas.

With respect to the behaviour of the control variables, the results are mixed. The impact of GDP per capita (in levels or squared values) is unclear when the full sample of countries is observed, but in developing countries of the Americas taken by themselves it seems to decrease income disparities initially but eventually worsen them. Greater urbanization seems, for the most part, to reduce income inequality in developing countries of the Americas, but the opposite seems to be true when Canada and the United States are included in the sample. In all cases, remittances and inflation decrease income inequality and unemployment and bigger government worsens it. Openness to trade seems to increase inequalities for the full sample of countries, but there are indications that it may mitigate it when developing countries alone are taken. Savings seem to decrease inequality in the full sample of countries and to worsen it when Canada and the United States are excluded. Internet access decreases inequality in developing countries of the Americas, but its impact on the full sample of countries is unclear. Finally, weaker political and civil rights seem to be more conducive to growth in developing countries of the Americas only, while the impact on the full sample of countries is indeterminate.

The most important policy implication of the findings presented here is that fiscal decentralization has so far fallen short of expectations in terms of its impact on growth, price stability and income distribution. Although there are some indications that decentralization can positively affect growth, particularly when it occurs on the expenditure side, the assumption was that it was going to be a powerful catalyst for this. Likewise, it was postulated that fiscal decentralization would act as a deterrent to fiscal mismanagement and hence would counteract inflationary pressures, but evidence for this has not yet materialized. Finally, it was presumed that this process would lead to greater income equality, and although there is evidence that this has started to happen (when the full sample of countries is considered and decentralization occurs on the revenue side), it also suggests that much work remains to be done before this goal is accomplished. The principal recommendation, particularly from the perspective of developing nations of the Americas, is that the institutional capacity of subnational levels of government. With greater institutional capacity and improved coordination between all layers of government, perhaps the positive outcomes anticipated for this process will begin to be realized.

# Bibliography

- Amagoh, F. and A. A. Amin (2012), "An examination of the impacts of fiscal decentralization on economic growth", *International Journal of Business Administration*, vol. 3, No. 6, Sciedu Press.
- Baskaran, T. and L. P. Feld (2013), "Fiscal decentralization and economic growth in OECD countries: is there a relationship?", *Public Finance Review*, vol. 41, No. 4, SAGE.
- Beramendi, P. (2003), "Political institutions and income inequality: the case of decentralization", WZB, Markets and Political Economy Working Paper, No. SP II 2003-09 [online] http://ssrn.com/abstract=473701.
- Blöchliger, H. (2013), "Decentralization and economic growth Part 1: how fiscal federalism affects long-term development", *OECD Working Papers on Fiscal Federalism*, No. 14, Paris, OECD Publishing [online] http://dx.doi.org/10.1787/5k4559gx1q8r-en.
- Bodman, P. and others (2009), "Fiscal decentralisation, macroeconomic conditions and economic growth in Australia", *Macroeconomics Research Group Working Paper*, No. 2609, University of Queensland [online] www.uq.edu.au/economics/mrg/2609.pdf.
- Boex, J. and others (2006), "Fighting Poverty through Fiscal Decentralization", Washington, D.C., United States Agency for International Development (USAID) [online] http://pdf.usaid.gov/pdf\_docs/PNADH105.pdf.
- Bojanic, A. N. (2016), "Fiscal decentralization, economic freedom, and political and civil liberties in the Americas", *Working Paper*, No. 1609, Tulane University [online] http://ideas.repec.org/p/tul/wpaper/1609.html.
- Brosio, G. and J. P. Jiménez (2013), *Decentralization and Reform in Latin America: Improving Intergovernmental Relations*, Northampton, Edward Elgar Publishing.
- Brueckner, J. K. (2005), "Fiscal federalism and economic growth", *CESifo Working Paper Series*, No. 1601 [online] www.econstor.edu/bitstream/10419/19065/1/cesifo1\_wp1601.pdf.
- Cavusoglu, T. and O. Dincer (2015), "Does decentralization reduce income inequality? Only in rich states", *Southern Economic Journal*, vol. 82, No. 1, Wiley.
- Davoodi, H. and H. Zou (1998), "Fiscal decentralization and economic growth: a cross-country study", *Journal of Urban Economics*, vol. 43, No. 2, Amsterdam, Elsevier.
- Durham, J. B. (1999), "Econometrics of income distribution: toward more comprehensive specification of institutional correlates", *Comparative Economic Studies*, vol. 41, No. 1, Springer.
- Gemmell, N., R. Kneller and I. Sanz (2013), "Fiscal decentralization and economic growth: spending versus revenue decentralization", *Economic Inquiry*, vol. 51, No. 4, Wiley.
- Goerl, C. and M. Seiferling (2014), "Income inequality, fiscal decentralization and transfer dependency", *IMF Working Paper*, No. 14/64, Washington, D.C., International Monetary Fund (IMF) [online] http://www.imf. org/external/pubs/ft/wp/2014/wp1464.pdf.
- Iqbal, N. and S. Nawaz (2010), "Fiscal decentralization and macroeconomic stability: theory and evidence from Pakistan", MPRA Paper, No. 27184 [online] http://mpra.ub.uni-muenchen.de/27184/.
- Islam, F. (2015), "The role of local self-government institution for deepening democracy at the grass-root level in Bangladesh", *Journal of Public Administration and Policy Research*, vol. 7, No. 2.
- Jalil, A. Z., M. Harun and S. H. Che Mat (2012), "Macroeconomic instability and fiscal decentralization: an empirical analysis", *Prague Economic Papers*, No. 2, Prague, University of Economics.
- Kaufman, H. (1969), "Administrative decentralization and political power", *Public Administration Review*, vol. 29, No. 1, Wiley.
- Kuznets, S. (1955), "Economic growth and income inequality", *American Economic Review*, vol. 45, No. 1, Nashville, Tennessee, American Economic Association.
- Lee, C. (2005), "Income inequality, democracy, and public sector size", *American Sociological Review*, vol. 70, No. 1, SAGE.
- Martínez, R., L. Ayala and J. Ruiz-Huerta (2001), "The impact of unemployment on inequality and poverty in OECD countries", *Economics of Transition*, vol. 9, No. 2, Wiley.
- Martínez-Vázquez, J. and R. M. McNab (2006), "Fiscal decentralization, macrostability, and growth", *Hacienda Pública Española/Revista de Economía Pública*, vol. 179, No. 4, Madrid, Instituto de Estudios Fiscales.
- \_\_\_\_(2003), "Fiscal decentralization and economic growth", *World Development*, vol. 31, No. 9, Amsterdam, Elsevier.
- Michels, A. (2011), "Innovations in democratic governance: how does citizen participation contribute to a better democracy?", *International Review of Administrative Sciences*, vol. 77, No. 2, SAGE.

- Monnin, P. (2014), "Inflation and income inequality in developed economies", *CEP Working Paper*, No. 2014/1, Council on Economic Policies [online] http://www.cepweb.org/wp-content/uploads/2014/05/CEP\_WP\_Inflation\_and\_Income\_Inequality.pdf.
- Moon, H. (2003), "Regional Inequality and Fiscal Decentralization in Korea: Evaluation and Policy Implications", Korea Development Institute [online] www.kdi.re.kr/upload/7050/IV\_HyungpyoMoon.pdf.
- Neyapti, B. (2010), "Fiscal decentralization and deficits: international evidence", *European Journal of Political Economy*, vol. 26, No. 2, Amsterdam, Elsevier.
- \_\_\_\_(2004), "Fiscal decentralization, central bank independence and inflation: a panel investigation", *Economic Letters*, vol. 82, No. 2, Amsterdam, Elsevier.
- Oates, W. (1993), "Fiscal decentralization and economic development", *National Tax Journal*, vol. 46, No. 2, National Tax Association.
- Okonkwo, O. N. and E. K. Godslove (2015), "Fiscal decentralization and Nigerian macroeconomic performance and economic stability", *International Journal of Economics and Finance*, vol. 7, No. 2, Canadian Center of Science and Education.
- Rao, M. G. (2000), "Fiscal Decentralization in Indian Federalism" [online] https://www.imf.org/external/ pubs/ft/seminar/2000/fiscal/rao.pdf.
- Rodden, J., G. Eskeland and J. Litvack (2003), *Fiscal Decentralization and the Challenge of Hard Budget Constraints*, Cambridge, Massachusetts, The MIT Press.
- Rodríguez, F. and D. Rodrik (1999), "Trade policy and economic growth: a skeptic's guide to cross-national evidence", *NBER Working Paper*, No. 7081, Cambridge, Massachusetts, National Bureau of Economic Research (NBER) [online] www.nber.org/papers/w7081.
- Rodríguez-Pose, A. and R. Ezcurra (2011), "Is fiscal decentralization harmful for economic growth? Evidence from the OECD countries", *Journal of Economic Geography*, vol. 11, No. 4, Oxford University Press.
- Sacchi, A. and S. Salotti (2014), "The effects of fiscal decentralization on household income inequality: some empirical evidence", *Spatial Economic Analysis*, vol. 9, No. 2, Taylor & Francis.
- Sepúlveda, C. and J. Martínez-Vázquez (2011), "The consequences of fiscal decentralization on poverty and income inequality", *Environment and Planning C: Government and Policy*, vol. 29, No. 2, London, Pion.
- Song, Y. (2013), "Rising Chinese regional income inequality: the role of fiscal decentralization", *China Economic Review*, vol. 27, Amsterdam, Elsevier.
- Thornton, J. (2007), "Fiscal decentralization and economic growth revisited", *Journal of Urban Economics*, vol. 61, No. 1, Amsterdam, Elsevier.
- Treisman, D. (2000), "Decentralization and inflation: commitment, collective action, or continuity?", *American Political Science Review*, vol. 94, No. 4, American Political Science Association.
- Weingast, B. R. (2009), "Second generation fiscal federalism: the implications of fiscal incentives", *Journal of Urban Economics*, vol. 65, No. 3, Amsterdam, Elsevier.
- Xie, D., H. Zou and H. Davoodi (1999), "Fiscal decentralization and economic growth in the United States", Journal of Urban Economics, vol. 45, No. 2, Amsterdam, Elsevier.
- Yifu Lin, J. and Z. Liu (2000), "Fiscal decentralization and economic growth in China", *Economic Development* and Cultural Change, vol. 49, No. 1, Chicago, The University of Chicago Press.